Author's response to reviews

Title: Inverse probability weighting and doubly robust methods in correcting the effects of non-response in the reimbursed medication and self-reported turnout estimates in the ATH survey

Authors:

Tommi Tommi Härkänen (tommi.harkanen@thl.fi)
Risto Kaikkonen (risto.kaikkonen@thl.fi)
Esa Virtala (esa.virtala@thl.fi)
Seppo Koskinen (seppo.koskinen@thl.fi)

Version: 2
Date: 14 May 2014

Author's response to reviews:

Line and page numbers in the manuscript. Page breaks removed.